**Author’s response to reviews**

**Title:** Maternal knowledge and attitudes towards complementary feeding in relation to timing of its initiation in rural Bangladesh

**Authors:**

Aatekah Owais (aowais@emory.edu)

Parminder Suchdev (psuchdev@emory.edu)

Benjamin Schwartz (bschwartz240@gmail.com)

David Kleinbaum (dkleinb@emory.edu)

A. S. G. Faruque (gfaruque@icdrrb.org)

Sumon Das (sumon@icdrrb.org)

Aryeh Stein (aryeh.stein@emory.edu)

**Version:** 2 **Date:** 11 Jan 2019

**Author’s response to reviews:**

Please see uploaded document that preserved formatting